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11 October 2009
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 41 of 2009, which covers the period from 04 through 10 October 2009.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 79.2 to 106.9 kcfs.  Daily average project tailwater ranged from 7.8’ to 10.0’ msl.  Daily average powerhouse forebay elevation ranged from 77.0’ to 77.5’ msl.  Daily average river temperature ranged from 63ºF to 65ºF.  Secchi disk measurements ranged from 6.0’ to 7.0’ all week.
1.2 Spill Bays 1 and 18, which provide attraction flow for Cascades Island and B-branch, respectively, are open 6” during daylight hours.  Daily average spill ranged between 1.4 to 1.5 kcfs.  

1.3 Unit Operation.  Powerhouse Two is the priority powerhouse.
1.3.1 Unit 3 was out of service from 1010 until 1103 on 08 October due to bank 3/4 tripping offline.

1.3.2 Unit 4 has remained out of service since 14 September 2009 for five-year overhaul.
1.3.3 Unit 7 has remained out of service since 04 June 2007 for turbine rehab.

1.3.4 Unit 9 has remained out of service since 04 August 2008 for turbine rehab.
1.3.5 Unit 11 has remained out of service since 12 June 2008 due to rotor damage.
1.3.6 Unit 17 went out of service at 1009 on 05 October for a four year overhaul.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris.
1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  No AWS valves were closed this reporting week: 

2.2 STS/VBS inspections.  The following STS/VBS inspections occurred this reporting week:
Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	1-2
	NA
	NA
	Scheduled for removal on 19 October.

	11, 16
	NA
	NA
	Out of service

	12-15, 17-18
	NA
	NA
	Scheduled for the week of 19 October.


2.2.1 Dewatering and Fish Salvage.  The following dewatering/fish salvage activities occurred this week:
2.2.1.1 On 05 October, Fisheries removed thousands of juvenile shad from U17 A gatewell slot.
2.2.1.2 On 06 October, Project Fisheries and Fisheries Field Unit removed 23 sturgeon, two catfish and five bullhead from Unit 17 draft tube.  Fish were returned to the river in good condition.

3 RESEARCH  

3.1 CRITFC - Annual chinook, coho, sockeye and steelhead sampling in the AFF continues.  
3.2 CRITFC- JMF video.  New video equipment installed for monitoring juvenile fish passage.  
3.3 NOAA- PIT tag detectors.  Vibration sensors remain installed on Spill bay 6. 

3.4 NWW truck release site.  Trucking and release at BON began on 16 August; continues every other day.

3.5 PSMFC- Smolt monitoring.  Daily condition and index monitoring continues.  Two steelhead and four salmon were observed crossing the separator bars.  Gas Bubble Trauma (GBT) exams occur twice a week; see www.fpc.org/CurrentDaily/Gbtsum.txt for further details.  
3.6 WDFW Fish Counting.  Visual counting continues at Bradford Island and Washington Shore fishways. 
4 FISHWAYS 

4.1 Project biologists inspected 05 through 10 October.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND (BI)  

	10/4
	All fishways
	No inspection
	Personnel out sick
	Control Room operators inspected.
	

	10/5
	A-branch weir
	Reading <1.0’>
	High forebay
	Self corrected
	

	CASCADES ISLAND (CI) nothing to report

	
	
	
	
	
	

	WASHINGTON SHORE (WS) 

	10/8, 10/10
	Weir 37
	Reading <1.0’
	Unknown
	No action taken.
	

	PH1 ITS – please see 4.5.6


4.2.1 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 09/10.  

4.2.2 Design of a mechanical velocity meter at the PH2 south monolith is in progress.  

4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island; the junction of the UMT and the Washington Shore ladder.  

4.3.1 Three out of four temperature probes are experiencing technical difficulties.
Table 3.  Bonneville Fishway Temperatures

	Date
	WS Avg
	BI Avg

	10/04
	NA
	63.3

	10/05
	NA
	62.8

	10/06
	NA
	62.1

	10/07
	NA
	61.8

	10/08
	NA
	62.0

	10/09
	NA
	61.5

	10/10
	NA
	60.8


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 09 October.
4.5.2 The Washington Shore fishway entrances remain in manual due to SA-24 board calibration issues.  

4.5.3 There are currently 11 of 12 floating orifice gates (FOG) installed at Powerhouse 2.  One gate was removed for maintenance and replaced with a bulkhead.
4.5.4 FV4-3 remains in manual due to a defective conduit sensor.  Repairs scheduled for 09/10.
4.5.5 DSM1 continues to head south.  STSs are in Units 1 and 2.  Orifice lights are on in those units as well.  Orifices are open for units 1-9.  
4.5.5.1 STSs and VBS will be pulled, and DSM1 dewatered, starting on 19 October to facilitate the ITS wall removal contract.

4.5.6 Chaingates are not quite to FPP criteria.  Automated gates 3B and 10B are manually set to 2.5’.  Automated gate 6C still is not working properly.  Chain gate 6B has been opened to maintain at least 2.5’ opening.  With the high forebay, 1B has been closed and 1C opened to 72’ msl.
4.6 Juvenile Fishways.
4.6.1 The B2CC remains closed.
4.7 Lamprey Fishways.  
4.7.1 The Cascades Is. entrance weir and bulkhead slot filler are installed.  
4.7.2 The CI LPS and PH2 NDE ramp are dewatered for the season.
4.7.3 The BI and WS AWS LPSs remain watered-up. 
5 CONSTRUCTION
5.1 PH1 Gantry 3.  Crane rails scheduled for replacement in FY10.  New crane in FY11.

5.2 PH1 Ice and Trash Sluiceway Wall Removal.  Wall and DSM1 outfall scheduled for removal in FY10.
5.2.1 The Program Logic Control (PLC) for the automated gates has not been completed.
5.3 TIE crane.  The crane is scheduled to be available by the 2010 fish passage season.
6 HAZMAT, SPILLS AND CLEANUP.   Nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 16 July changes PH2 unit priority to: 12,18,15,17,14,13,16.  This change is effective from 1100 on 16 July until U11 returns to service.
7.2 A teletype sent on 27 July changes the unit operation for PH2.  PH2 units will operate at the low end to mid point of the 1% efficiency range.

7.3 A teletype sent on 01 September requests BON forebay maintain 71.5’ to 78.0’ msl as a hard constraint and 71.5’ to 77.5’ as a soft constraint from 0001 on 08 September 2009 until 31 March 2010.  
7.3.1 From 07 November through 31 December, 78.0’ msl will occur as needed to support chum spawning below the dam.  
7.3.2 This requested operation is a deviation from the Bonneville Water Control Manual.  The deviation was coordinated with the Region on 21 August.
7.3.3 This operation is requested to facilitate TDA spillwall work.  If the spillwall construction is completed early, a return to normal forebay operating elevations will be coordinated. 

7.4 A teletype sent on 08 September requests that BON operate the single highest priority unit at PH1 (currently Unit 1) as needed.  This is to reduce the risk of BON going outside the 1% efficiency range while supporting the high forebay request for TDA spillwall work.  This request is effective immediately until further notice.

JERRY CARROLL
Acting Operations Manager
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